High-latitude ocean habitats are a crucible of fish body shape diversification

Michael D. Burns^{1,}, Sarah T. Friedman^{1,}, Katherine A. Corn^{2,}, Olivier Larouche^{3,}, Samantha A. Price^{4,}, Peter C. Wainwright^{1,}, Edward D. Burress^{5,}

¹Department of Evolution & Ecology, University of California, Davis, Davis, CA, United States

²Department of Biological Sciences, Virginia Polytechnic Institute & State University, Blacksburg, VA, United States

³Department of Biology and Biochemistry, University of Houston, Houston, TX, United States

⁴Department of Biological Sciences, Clemson University, Clemson, SC, United States

⁵Department of Biological Sciences, University of Alabama, Tuscaloosa, AL, United States

Corresponding author: Department of Evolution & Ecology, University of California, Davis, Davis, CA 95616, United States. Email: burnsmic01@gmail.com

Abstract

A decline in diversity from the equator to the poles is a common feature of Earth's biodiversity. Here, we examine body shape diversity in marine fishes across latitudes and explore the role of time and evolutionary rate in explaining the diversity gradient. Marine fishes' occupation of upper latitude environments has increased substantially over the last 80 million years. Fishes in the highest latitudes exhibit twice the rate of body shape evolution and one and a third times the disparity compared to equatorial latitudes. The faster evolution of body shape may be a response to increased ecological opportunity in polar and subpolar oceans due to (1) the evolution of antifreeze proteins allowing certain lineages to invade regions of cold water, (2) environmental disturbances driven by cyclical warming and cooling in high latitudes, and (3) rapid transitions across depth gradients. Our results add to growing evidence that evolutionary rates are often faster at temperate, not tropical, latitudes.

Keywords: latitudinal diversity gradient, ecological opportunity, evolutionary rates, MuSSCRat, disparity, linear morphometrics

Lay Summary

Regions of Earth that are closer to the tropics tend to have more biodiversity than regions that are closer to the poles. We explored this pattern by examining body shape diversity across marine fishes, a group of aquatic animals that span latitudes from tropical waters at the equator to polar and subpolar oceans at the poles. We found that modern marine fishes moved into polar and subpolar oceans relatively recently, with the invasion of these oceans occurring within the last 80 million years. Contrary to the common pattern, marine fishes' body shape diversity is higher in high temperate regions than in tropical regions. Underlying this pattern, we find that polar and subpolar marine fishes evolve body shapes faster than species in the tropics. The more rapid evolution of body shape in temperate species may be due to the evolution of unique antifreeze proteins that have allowed polar and subpolar fishes to exist in icy seawater, cyclical changes in the climate at polar and subpolar latitudes, and a higher rate of transition across depth habitats. Despite the typical biodiversity pattern, our results add to growing evidence that evolutionary rates are often faster at temperate, not tropical, latitudes.

Introduction

The latitudinal diversity gradient, seen as a strong decline in diversity from the equator to the poles, is one of Earth's most dominant biodiversity patterns. Although studies on the latitudinal diversity gradient have traditionally focused on patterns of species richness, recent research has also looked at how trait diversity varies with latitude as another dimension of diversity (Drury et al., 2021; Freeman et al., 2022; Wiens, 2023). One hypothesis to explain the greater diversity in the tropics includes greater time for species and morphological diversity to evolve due to a more stable biome (Fischer, 1960; Hawkins et al., 2006; Mittelbach et al., 2007; Wiens & Donoghue, 2004). Other hypotheses predict that faster diversification rates underlie the increase in diversity, either because higher temperatures increase metabolism, which accelerates genetic divergence (Salisbury et al., 2012; Schluter,

2016; Schluter & Pennell, 2017), or because there is greater ecological opportunity, with many niche combinations due to more energy, annual productivity, and biotic interactions (Etienne et al., 2019; Fenton et al., 2023; Pigot et al., 2016; Schluter, 2016). Thus, on the one hand, evolutionary dynamics may be relatively muted, but the stability of tropical regions has allowed lineages time to accumulate diversity, while on the other hand, the tropics are seen as a more ecologically productive and dynamic place that supports higher rates of diversification. Studies on a wide range of taxa have supported both perspectives (Egan et al., 2022; Fischer, 1960; Hawkins et al., 2006; Mittelbach et al., 2007; Schluter, 2016; Schluter & Pennell, 2017).

Although explanations of latitudinal diversity gradients focus on faster evolutionary rates at lower latitudes, others have hypothesized that higher rates should occur in areas with low

Received November 2, 2023; revisions received April 12, 2024; accepted April 18, 2024

[©] The Author(s) 2024. Published by Oxford University Press on behalf of The Society for the Study of Evolution (SSE) and European Society for Evolutionary Biology (ESEN).

This is an Open Access article distributed under the terms of the Creative Commons Attribution License (https://creativecommons.org/licenses/by/4.0/), which permits unrestricted reuse, distribution, and reproduction in any medium, provided the original work is properly cited.

species diversity (Freeman et al., 2022; Lawson & Weir, 2014; Schluter, 2016; Schluter & Pennell, 2017). Evolution may be the fastest where species diversity is lowest because increased biotic interactions in species-rich regions may limit ecological opportunity and constrain divergence (Freeman et al., 2022; Lawson & Weir, 2014; Schluter, 2016; Schluter & Pennell, 2017). Under this scenario, morphological and speciation rates should be fastest in species-poor temperate zones and slowest in species-rich tropical zones. In support of these predictions, studies across diverse groups have shown that speciation rates and rates of trait diversification are substantially faster in temperate latitudes (Freeman et al., 2022; Lawson & Weir, 2014; Schluter & Pennell, 2017; Weir & Wheatcroft, 2010). While not initially intuitive, this pattern of high speciation and trait evolution rates in relatively depauperate high-latitude areas raises the possibility that these regions, precisely because they have fewer species and biotic interactions, offer conditions that favored rapid adaptive diversification in the recent past (Rabosky et al., 2018; Schluter & Pennell, 2017).

Marine fishes are a particularly interesting group concerning the latitudinal diversity gradient. They strongly exhibit the expected pattern in species richness (Rabosky et al., 2018), with vastly more species in the tropics than at high latitudes. However, speciation rates are significantly higher in high-latitude lineages (Rabosky et al., 2018), suggesting that ecological opportunity might be higher in temperate latitudes. Moreover, a recent study associated high speciation rates in high latitudes with repeated diversification along a depth gradient (Friedman & Muñoz, 2023). Elevated speciation rates in high-latitude marine fishes might also reflect a recent and ongoing poleward expansion of marine diversity (Miller et al., 2018; Rabosky et al., 2018), one which includes some notable adaptive radiations, such as icefishes and rockfishes (Kolora et al., 2021; Near et al., 2012).

In the present study, we explore the nature of marine fish phenotypic diversification across latitudes. Using a phylogenetic approach, we estimate the history of latitudinal occupation by marine teleost fishes and then characterize latitudinal gradients in body shape diversity and the rate of body shape evolution. The expectation and general impression that tropical regions contain greater phenotypic diversity is common in the literature (Rabosky et al., 2018; Ricklefs, 2012) but is inconsistently observed in empirical studies of various groups (Hipsley et al., 2014; Roy & Martien, 2001; Stuart-Smith et al., 2013). These conflicting results indicate that the relationship between species richness and phenotypic diversity is complex, and areas of high species richness are not always more morphologically diverse (Hipsley et al., 2014; Ricklefs, 2012; Roy & Martien, 2001).

Coral reefs and other tropical marine habitats contain many fish species (Rabosky et al., 2018; Stuart-Smith et al., 2013) and high amounts of phenotypic diversity (Price et al., 2019), reflecting adaptations to habitat (Friedman et al., 2020; Price et al., 2013), trophic ecology (Corn et al., 2022; Friedman et al., 2016), and mode of locomotion (Friedman et al., 2021; Larouche et al., 2020). Nevertheless, temperate lineages have also evolved notable morphological variation despite lower species richness. Some temperate lineages exhibit diversification dynamics linked to increased ecological opportunity due to fluctuations in nearshore habitats during oscillations in global temperature (Eastman, 1991; Ingram, 2015; Parker et al., 2022). Furthermore, polar oceans and the temperate North Pacific are home to several radiations that display elevated rates of speciation and morphological evolution (Ingram, 2015; Near et al., 2012; Parker et al., 2022). Thus, both tropical and temperate latitudes may facilitate exceptional radiations of form, but the overall trend in body shape diversity is unclear.

We assess the latitudinal gradient in body shape diversity across 3,183 species of marine fishes. We ask whether differences in time for morphological evolution or rates of evolution provide a better explanation for the observed gradient in body shape diversity. We examine how latitudinal occupation varied through time and whether assemblages reveal a latitudinal diversity gradient. We examine the underlying cause of the latitudinal gradient in body shape diversity in the context of the history of occupation of the latitude quartiles and the rate of body shape evolution in each region. Our macroevolutionary study shows that marine fishes have rapidly expanded their occupation of high-latitude regions in the past 80 million years and experienced a coincident acceleration of body shape evolution. Together with the previously shown elevated speciation rates, our results suggest a strong response to elevated ecological opportunity throughout the Cenozoic.

Methods

Data acquisition

Body shape data for marine teleost fishes were taken from a previously published collection of measurements made from museum specimens (Price et al., 2019, 2022). We trimmed this dataset to 3,183 species that matched a tip in a recent time-calibrated phylogenetic hypothesis for ray-finned fishes (Rabosky et al., 2018) retrieved using the fishtree R package (Chang et al., 2019). We analyzed species averages of eight linear measurements that encapsulate the three dimensions of body shape, including standard length, jaw length, mouth width, body width and depth, caudal peduncle width and depth, and head depth. We log-transformed and size-corrected the linear traits by taking the residuals of phylogenetic regressions on body size implemented with the phyl. resid function in the R package phytools (Revell, 2012). We used the geometric mean of the three major axes of body shape (standard length, maximum body width and maximum body depth) as a proxy for body size (Price et al., 2019).

Latitude range centroids were taken from the (Rabosky et al., 2018) study and, following that study, were grouped into discrete latitude bins of the same number of species based on absolute latitude quartiles. This approach allowed us to have equal taxon sampling between the discrete latitude bins. Data available from the Dryad Digital Repository: https://doi.org/10.5061/dryad. fc71cp4. The lower quartile represented species between 0° and 2.51°, the second quartile between 2.52° and 11.50°, the third quartile between 11.51° and 29.72°, and the upper quartile were species above 29.72° latitude (see Supplementary Table S1).

Reconstructing latitudinal occupation

To reconstruct the history of latitude occupation, we estimated ancestral states with the R package corHMM, using joint inferences of the ancestral states. We fit alternative versions of the Mk model, one model with equal rates (ER) and another with all rates different, and evaluated the presence of one, two, three, or four rate regimes per state across branches (Beaulieu et al., 2013). We used an unrestricted CTMC that permitted transitions between all quartiles to accommodate cladogenic changes in quartile habitation. Transitions between nonadjacent quartiles were permitted in our dataset as taxa were assigned to quartiles based on the centroid of their occupancy range, and thus, the range of a species may extend beyond the quartile to which it is assigned. Those models were compared using sample size corrected Akaike information criteria (AICc) and Akaike weights (AICw). We then used the best-fit model parameters (Supplementary Table S2) to generate a distribution of 1,000 stochastic character maps using make-Simmap (Beaulieu et al., 2013). We then calculated the proportion of branches at million-year intervals in each latitude quartile to determine the proportion of time spent in each state throughout the last 190 million years. This approach does not consider historical biogeographic processes that may have influenced the midrange latitude occupation of marine fishes, including the centers of origin for extant lineages or how ranges have shifted during continental drift.

Estimating latitudinal gradients of morphospace occupation, evenness, and disparity

To determine the primary axes of morphological diversity in marine fishes, we performed a principal components analysis (PCA). We calculated body shape disparity for the assemblage of species found in each latitude quartile as body shape variance using the function morphol.disparity in the R package geomorph (Baken et al., 2021). Disparity was calculated across all axes using the overall mean. We computed the multivariate (matrix of all eight morphological traits) morphological variance for each quartile. To visualize how patterns of disparity varied geographically, we combined morphological disparity with species occurrence data collected by Rabosky et al. (2018). The occurrence data is constructed as a global grid with 150 km \times 150 km resolution and details the species present in each cell. We determined the species in our dataset present in each cell and evaluated the disparity for every marine cell for 22,736 cells. These results were then plotted as a heatmap across the globe.

We estimated the extent to which fish in each quartile have filled the morphospace by computing the volume of morphospace occupation in each latitude quartile as the multivariate volume of the minimum polygon. We used the first six axes of the PCA, which accounted for 98.2% of the variance. We then calculated the absolute difference in volume between each pair of quartiles. To assess whether the differences in volume between quartiles were statistically significant, we permuted quartile assignments among our species data 1,000 times and compared the volume among the permutated quartiles. We calculated the number of replicates in which the permutated difference in volume was greater than the empirical difference in volume. Finally, we divided this number by the total number of replicates to determine whether the differences in the empirical volume between quartiles were significant.

We estimated the functional evenness to describe the regularity of the distribution of species (and their abundances) in trait space (Villéger et al., 2008) using the fundiversityR package (Grenié & Gruson, 2023). We used the first six axes of the PCA to calculate the overall functional evenness, as well as each principal component individually. We then calculated the difference in functional evenness between each pair of quartiles. To assess whether the differences in functional evenness between quartiles were statistically significant, we permuted quartile assignments among our species data 1,000 times and compared the functional evenness among the permutated quartiles. We then calculated the number of iterations in which the permutated difference in functional evenness was greater than the empirical difference in functional evenness. Finally, we divided this number by the total number of iterations to determine whether the differences in the empirical functional evenness between quartiles were significant.

The dispersion of taxa in a morphospace can be biased by how many taxa are sampled within each clade and which specific taxa are measured from those clades. To determine whether the various species sampled from each clade might be biasing our estimates of morphological variance, volume, and evenness, we performed a bootstrap analysis 1,000 times for the different dispersion metrics. We qualitatively compared the distribution of bootstrapped values to the empirical values for each analysis. Bootstrapping the empirical data allows us to determine whether the different analyses of morphological dispersion are robust to any potential bias in taxon sampling.

Evolutionary rates

To evaluate rates of body shape evolution across the latitudinal gradient, we employed a modified rate-by-state test (Reynolds et al., 2016). First, we estimated tip rates (i.e., species-specific rates) using a relaxed-clock multivariate model of evolution following (Burress et al., 2020), implemented in RevBayes (Höhna et al., 2016). We used a random local clock model to estimate the rates. The Markov chain Monte Carlo (MCMC) was run for 300,000 iterations with 10% burnin. We then regressed the tip rates against latitude and absolute latitude midpoints. We assessed the correlation between tip rates and mid-range latitude using a phylogenetic regression with robust M-estimation (Revell, 2010). We used a robust M-estimation because it is less sensitive to outliers, and our tip rates contain outlier species. The robust phylogenetic regression was done using the R package ROBRT (Adams et al., 2023). Furthermore, we plotted the geographic distribution of tip rates using the occurrence data collected by Rabosky et al. (2018) following the methodology outlined above. Tip rates for each geographic region were calculated as the average tip rate of the species inhabiting each 150 km × 150 km grid.

To further test the relationship between rates of body shape evolution and latitude, we employed a state-dependent, uncorrelated log-normal (UCLN) model of evolution (Drummond et al., 2006; May & Moore, 2020), implemented in RevBayes (Höhna et al., 2016). The MuSSCRat model jointly estimates the evolutionary history of the discrete and continuous characters and accommodates background rate variation (i.e., rate heterogeneity not attributable to the focal discrete character), thereby reducing biases in rate estimates (Burress & Muñoz, 2022; May & Moore, 2020; Revell, 2013). As a discrete character, we used latitudinal quartiles as described above, and as continuous characters, we used the eight size-corrected body shape traits. The MCMC was run for 300,000 iterations with 10% burnin. To evaluate the sensitivity of posterior parameter estimates, we repeated the MCMC across different priors on the number of rate shifts (400, 500, and 600 shifts).

Results

Evolutionary history of latitudinal occupation

The stochastic character maps indicate a relatively stable occupation of all latitude quartiles from the base of the phylogeny, about 190 Mya until about 80 Mya, when the upper quartile (including polar, subpolar, and temperate fishes) started to increase in proportion (Figure 1). Prior to about 80 Mya, the lower, second, and third quartiles were occupied by about 90% of extant marine teleost lineages, with only about 10% of lineages occurring within the upper quartile. Beginning shortly before the end-Cretaceous, the proportion of lineages occupying upper quartile latitudes steadily increased until reaching its peak today at about 25% of branch length in the past million years.



Figure 1 A bar plot showing the proportion of branches at million-year intervals in each absolute latitude quartile generated from the ancestral state reconstruction using stochastic character mapping. The three rate categories in each quartile correspond to hidden rate categories, ranging from the lowest transition rate (Rate 1) to the highest (Rate 3). The lower quartile represented species between 0° and 2.51°, the second quartile between 2.52° and 11.50°, the third quartile between 11.51° and 29.72°, and the upper quartile were species above 29.72° (see inset map for quartile locations).

Body shape evolution and disparity across latitude

Body shape varies substantially in marine fishes (Figure 2). Species in each latitude quartile are found in all regions of the morphospace, with marine fishes evolving shapes that range from highly elongate bodies, those that are dorsoventrally compressed, and others that are laterally compressed, regardless of latitude. The volume of morphospace occupied differed slightly between the quartiles, but only the upper and second quartiles exhibited significant differences (Supplementary Figure S1 and Supplementary Table S3). The volume of morphospace occupied by each quartile was similar when the data were bootstrapped (Supplementary Figure S2). The distribution of species occupying each quartile showed qualitative variation. Species in the lower, second, and third latitude quartiles are concentrated in an oval-shaped region of morphospace distributed in the lower half of PC1 and PC2, represented by mostly laterally compressed, deep body shapes. While species in these latitudes evolved into all other regions of morphospace, these body shapes are represented by fewer species. In contrast to the other latitude quartiles, body shapes of upper quartile species are significantly more evenly distributed across Principal Component 1 (Supplementary Table S4), with similar results when the data was bootstrapped (Supplementary Figure S3). Most notable is the concentration of species with elongated, somewhat wide body shapes in the upper quartile, a morphotype that is more poorly represented in the other quartiles (Figure 2).

These impressions of distributions within morphospace were confirmed by the disparity of multivariate body shape, which was highest in the upper quartile compared to the other latitude quartiles (Table 1), though significant differences only occurred between the upper and lower quartiles. No other latitude quartile showed significant differences in body shape disparity, although the trend is for increasing disparity as one moves away from the equator (Table 1; Figure 3). However, when disparity is viewed on a finer geographic scale, we see variation in patterns of disparity within each latitudinal quartile (Figure 3). Some regions close to the equator had high levels of disparity, including the west coast of Africa and the eastern portion of the Neotropics (Figure 3), while some regions near the poles had low disparity. When bootstrapping the data, we see that body shape disparity was consistently higher in the upper quartile compared to the other latitude quartiles (Supplementary Figure S4). These results suggest that the effect of latitude on body shape evolution is particularly acute in the upper quartile.

Rates of body shape evolution across latitude

Rates of body shape evolution varied among marine fishes (~31fold), including elevated rates in eelpouts (Zoarcidae), icefishes and their allies (Notothenioids), and cods (Gadidae) (Figure 4A). A phylogenetic regression shows that the rates were significantly but weakly correlated with latitude (Figure 4B; $R^2 = 0.13$; p < 0.05) and absolute latitude (Figure 4C; $R^2 = 0.12$; p < 0.05). The rate of body shape evolution was state-dependent (Posterior Probability > 0.95), with the fastest rates occurring in the highest latitudes (Figures 3 and 4D). This result was consistent across replicates with different priors (all P p > 0.95). The upper latitude quartile had faster evolutionary rates than the lower (2.1-fold), second (1.9-fold), and third (1.9-fold) quartiles (Figure 3D).

Discussion

Body shape varies spectacularly in marine fishes, but the diversity of body shapes is not spread evenly between the equator and the poles. Although the tropical lower latitudes house vastly more species, we find that the disparity of body shape is greatest and is evolving fastest at mid to high latitudes. We also show that the occupation of upper-latitude oceans has increased substantially in the past 80 million years. Together with the dramatic environmental changes accompanying the cyclical warming and cooling that polar oceans undergo, these results suggest that the evolution of fishes in these regions has fluctuated throughout the



Figure 2 Major axes of body shape variation in 3,183 species of marine teleost fishes, represented by the first two principal component axes (PCs) from a principal components analysis on eight linear traits. Each point is the average shape of a species, colored by the absolute latitude quartile. Drawings show representative species illustrating morphological variation across the plot. The lower quartile represented species between 0° and 2.51°, the second quartile between 2.52° and 11.50°, the third quartile between 11.51° and 29.72°, and the upper quartile were species above 29.72° (see inset map for quartile locations).

Table 1 Morphological variance and pairwise differences	
between variances for a multivariate matrix of all eight traits fo	١Ì
each latitude quartile.	

	Lower	Second	Third	Upper		
Procrustes variance for each group						
Procrustes variance	0.276	0.312	0.322	0.349		
Pairwise absolute diff	erence betv	veen varianc	es			
Lower Second Third Upper	-	0.036 -	0.046 0.009 -	0.073* 0.037 0.027 –		

Note. Asterisks represent significant differences at p < 0.05. The lower quartile represented species between 0° and 2.51°, the second quartile between 2.52° and 11.50°, the third quartile between 11.51° and 29.72°, and the upper quartile were species above 29.72°.

Cenozoic. Our results add to the growing body of evidence that the poleward expansion of organisms during Earth's most recent deglaciation has often led to more rapid evolution in the temperate zone than in the tropics.

Occupation of upper latitudes by extant marine lineages was relatively rare until an ongoing invasion that began in the late Cretaceous. Upper latitude occupation began increasing around 80 Mya, a period when many major lineages of marine teleosts originated (Ghezelayagh et al., 2022). While our ancestral state reconstructions do not account for historical biogeographic processes such as changes in the center of origins or continental drift, our finding of a recent increase in upper latitude occupation is supported by a previous biogeographical analysis which found that the colonization of high latitude marine habitats increased over the Cenozoic (Miller et al., 2018). This relatively recent history of upper latitude fishes indicates that increased time for morphological evolution is not the driving factor behind their high disparity, and low rates of evolution are not the primary factor behind low disparity among fishes at the poles. Instead, we found that the rate of body shape evolution is substantially faster in latitudes closer to the poles. Rates decrease substantially as species move closer to the Earth's equator (Figure 3). Increased upper latitude occupation and elevated rates of body shape evolution occur in several major North Pacific and Polar Ocean clades (Figure 1), including the eelpouts and icefishes (Figure 3). Together with previous work (Miller et al., 2018; Rabosky et al., 2018), these results demonstrate that while tropical regions house the greatest marine teleost species diversity, lineages in upper temperate regions are currently experiencing more rapid body shape evolution and speciation.

One classic hypothesis for the latitudinal diversity gradient is that greater ecological opportunity in the tropics, caused by increased productivity and biotic interactions, drives higher rates



Figure 3 Mean tip rate and disparity of body shape for marine fish assemblages at the global scale. Cell colors correspond to the scale bar, with warmer colors representing higher rates of morphological evolution and morphological disparity. Grid cell size is 150 km × 150 km for both images.



Figure 4 (A) Branch-specific rates of morphological evolution across a phylogenetic tree of marine teleost fishes. Warmer colors represent faster rates of evolution. Colored circles at the tip represent latitude quartiles. There is a significant relationship between morphological evolutionary rate, (B) latitude, and (C) absolute latitude. (D) Density plot of state-dependent rates of body shape evolution estimated with a Bayesian, state-dependent, relaxed-clock model of multivariate Brownian motion. The lower quartile represented species between 0° and 2.51°, the second quartile between 2.52° and 11.50°, the third quartile between 11.51° and 29.72°, and the upper quartile were species above 29.72° (see inset map for quartile locations).

of species and morphological diversification through character displacement and niche partitioning (Etienne et al., 2019; Fenton et al., 2023; Pigot et al., 2016; Schluter, 2016). In contrast to this viewpoint, others have predicted that ecological opportunity is highest in areas with low species diversity because the decreased biotic interactions result in more open niches (Cooper & Purvis, 2010; Freeman et al., 2022; Weir & Wheatcroft, 2010). Ecological opportunity can increase through the availability of new adaptive zones (Simpson, 1944) following key innovations or entry into new geographic regions with unfilled niches (Givnish, 2015; Schluter, 2000; Simpson, 1953), or a combination of the two (Near et al., 2012; Schluter, 2000). Both mechanisms potentially operated in upper latitude marine fishes. The two groups with the highest rates of phenotypic evolution, the eelpouts and icefishes (Figure 3), contain lineages that independently evolved antifreeze proteins (Deng et al., 2010; Hew et al., 1988; Near et al., 2012). This novelty allowed these groups to exist in seawater that can be a full degree lower than the freezing point of their body fluids, as found in polar and subpolar oceans (Bar Dolev et al., 2016; Fletcher et al., 1986; Raymond & DeVries, 1977) that otherwise contain relatively few other lineages (Eastman, 1991; Hobbs et al., 2020). The evolution of antifreeze proteins allowed these lineages to invade a biome rich in open niches independently and has increased the rates of morphological evolution and speciation in notothenioid icefishes (Dornburg et al., 2017; Near et al., 2012). Furthermore, the extinction of other competitors enabled these ancestrally benthic fishes to invade newly available pelagic foraging niches, resulting in associated morphological diversification (Eastman, 2005; Eastman & DeVries, 1981; Matschiner et al., 2011). However, other upper latitude clades have evolved antifreeze proteins and do not exhibit elevated body shape evolution rates, including snailfishes and sculpins (Fletcher et al., 2001; Gong et al., 1995; Wainwright & Longo, 2017). Thus, the evolution of antifreeze proteins that facilitated access to polar and subpolar oceans in upper latitudes has likely played a role in the high rates of body shape evolution seen in some of the upper latitude lineages, though it is not the only factor influencing the pattern.

Ecological opportunity in polar and subpolar oceans in upper latitudes also may have been impacted by cyclical warming and cooling, resulting in environmental disturbances and reshaping the physical environment across different geologic periods (Clarke & Crame, 2010; Dornburg et al., 2017; Near et al., 2012; O'Loughlin et al., 2011). The creation of novel niches due to environmental disturbance could underlie elevated rates of body shape evolution (Corn et al., 2022; Dornburg et al., 2017; Folk et al., 2019; Lucek et al., 2018; Near et al., 2012; Parker et al., 2022; Rutschmann et al., 2011). For instance, the periodic disturbance of the environment around Antarctica, in which ice scours eliminated habitat during cool periods and habitats reopened during warm periods, has been identified as a key factor behind the repeated ecomorphological diversification in cryonotothenioid icefishes (Parker et al., 2022) and eelpouts (Hotaling et al., 2021). The changing environment in upper latitude biomes may have facilitated the repeated generation of ecological opportunity and promoted the rapid evolution of body shape (Dornburg et al., 2017; Near et al., 2012; Parker et al., 2022; Rutschmann et al., 2011).

Many upper latitude lineages transition across depth gradients more rapidly than lower latitudes (Friedman & Muñoz, 2023), which may increase ecological opportunity and subsequent morphological diversification. For example, eelpouts (Zoarcidae), rockfishes (Sebastidae), flatfishes (Pleuronectidae), and icefishes and their allies (Notothenioids) transition across depth zones more readily than expected based on clade age and species richness (Friedman & Muñoz, 2023). In our dataset, these same clades exhibit some of the highest rates of body shape evolution. Different depths in marine environments provide multiple axes along which fishes can diversify, including diet, microhabitat, and life history strategies (Dornburg et al., 2017; Gerringer et al., 2021; Kolora et al., 2021). These different axes likely increase the potential for morphological evolution in these clades. For instance, Antarctic ice fishes diversify along the benthic-pelagic axis, with repeated ecomorphological diversification during invasions of pelagic, benthic, semipelagic, and epibenthic niches and exhibit subsequent morphological adaptation to dietary niches in those habitats (Parker et al., 2022; Rutschmann et al., 2011). Furthermore, temperate North Pacific Rockfish exhibit significant morphological diversification related to transitions across depth gradients and diet niches (Ingram, 2015; Ingram & Kai, 2014; Ingram & Shurin, 2009). Repeated diversification across depth gradients in upper latitude biomes may have contributed to the high disparity and rapid evolution of body shape.

Although we might predict that body shape diversity in marine fishes would be higher in tropical lower latitudes following the patterns of species richness instead, we find that the range of body shapes is relatively constant across latitudes while disparity increases in high temperate areas (Figure 2; Supplementary Figure S1). Despite some regions of high disparity in the tropics, often associated with continental edges such as the west coast of Africa and the eastern portion of the Neotropics (Figure 3), the overall pattern of low tropical body shape disparity and high species diversity exists because lower latitude species are more densely packed in morphospace. Species from tropical lower latitude habitats are particularly densely packed in a region of morphospace corresponding to laterally compressed, short, deep-bodied phenotypes (Corn et al., 2022). Coral reefs also house many fusiform or highly elongated species and a few that are very wide-bodied. These forms provide a variety of adaptations to life within highly structured environments (Domenici et al., 2008; Larouche et al., 2020; Webb, 1983) that facilitate precise maneuverability, moving within the sand or inside reef structures, and sustained swimming (Webb, 1982, 1984, 1988). In contrast, upper-latitude fishes often occur in the lower right region of the body shape morphospace (Figure 2). This includes several lineages that have long slender bodies with tapered tails, such as eelpouts (Zoarcidae), snailfishes (Liparidae), and grenadiers (Macrouridae), a body shape shown to perform efficient, slow-speed undulatory swimming (Tytell & Lauder, 2004; van Ginneken et al., 2005) and is also well-represented in the cold waters of the deep ocean (Martinez et al., 2021; Neat & Campbell, 2013). The laterally compressed, deep-bodied shape of actively foraging butterflyfishes, surgeonfishes, and other coral reef groups is much less common at upper latitudes. Instead, there is a greater occurrence of wide-bodied species, indicative of a benthic lifestyle, and the relatively wide, elongated forms that characterize efficient, slow swimming (Figure 2). Thus, water temperature may have pervasive effects on the activity of marine fishes, which at upper latitudes may manifest in more frequent use of body shapes that are adaptive for less active lifestyles due to decreasing metabolic rates and locomotor capacity in colder waters (Castro-Santos et al., 2022).

Some lineages of marine fishes exhibit antitropical distributions where they inhabit temperate or polar habitat patches on either side of the tropics (Briggs, 1987; Hubbs, 1952; Ludt, 2021; Ludt & Myers, 2021). These antitropical distributions have likely influenced patterns of diversification across a latitudinal diversity gradient within a few clades of ray-finned fishes. However, only 35 families of marine ray-finned fishes contain at least one antitropical species (see Hubbs (1952) and Randall (1981) for a review of families that exhibit an antitropical distribution), and the low number of species showing this pattern means antitropical distributions are unlikely to influence the broad patterns observed in our study. However, future studies should explicitly test whether antitropical distributions affect patterns of diversification in marine fishes.

Our study adds to evidence that recent evolutionary rates are often more rapid at upper latitudes than in lower latitudes (Freeman et al., 2022; Lawson & Weir, 2014; Miller et al., 2018; Rabosky et al., 2018; Schluter & Pennell, 2017; Weir & Wheatcroft, 2010). We show that extant marine fishes exhibit faster body shape evolution in upper versus lower latitudes, possibly because of increased ecological opportunity due to open niches, changing environments, and increased transitions across depth gradients. The faster evolution seen in upper latitudes across multiple taxonomic groups might be a shared evolutionary response to ecological opportunities created by the poleward expansion of organisms during Earth's cyclical warming (Miller et al., 2018; Schluter & Pennell, 2017). The recent rapid evolution at upper latitudes could continue as more lineages expand into warming temperate zones during accelerated climate change. However, the generality of the morphological pattern remains unknown as most studies have focused exclusively on speciation rates or looked at morphological evolution across only a few radiations. Until more studies examine how morphological diversification dynamics vary across latitudes, it will remain unclear whether the patterns uncovered here are general for most of life on Earth or unique to a few exceptional radiations.

Supplementary material

Supplementary material is available online at Evolution Letters.

Data and code availability

Morphological data have been deposited in Dryad (https://doi.org/10.5061/dryad.34tmpg4qf) and scripts are available on Zenodo (https://doi.org/10.5281/zenodo.7803539).

Author contributions

E.D.B. and P.C.W. designed the study. S.T.F., K.A.C., O.L., S.A.P., and P.C.W. developed the body shape data set. M.D.B. analyzed the data and wrote the manuscript. All authors contributed substantially to the preparation of the manuscript.

Funding

NSF grant DEB-1830127/1556953 to S.A.P. and P.C.W.

Conflict of interest: The authors declare no conflicts of interest.

Acknowledgments

M.D.B. thanks the members of the laboratory of P.C.W. for valuable discussions throughout the writing of this manuscript and Bill Ludt for helpful discussions on antitropical distributions in marine fishes.

References

- Adams, R., Cain, Z., Assis, R., & DeGiorgio, M. (2023). Robust phylogenetic regression. Systematic Biology, syad070. https://doi. org/10.1093/sysbio/syad070
- Baken, E. K., Collyer, M. L., Kaliontzopoulou, A., & Adams, D. C. (2021). geomorph v4.0 and gmShiny: Enhanced analytics and a new graphical interface for a comprehensive morphometric experience. Methods in Ecology and Evolution, 12(12), 2355–2363. https:// doi.org/10.1111/2041-210x.13723

- Bar Dolev, M., Braslavsky, I., & Davies, P. L. (2016). Ice-binding proteins and their function. Annual Review of Biochemistry, 85, 515– 542. https://doi.org/10.1146/annurev-biochem-060815-014546
- Beaulieu, J. M., O'Meara, B. C., & Donoghue, M. J. (2013). Identifying hidden rate changes in the evolution of a binary morphological character: The evolution of plant habit in Campanulid angiosperms. Systematic Biology, 62(5), 725–737. https://doi. org/10.1093/sysbio/syt034
- Briggs, J. C. (1987). Antitropical distribution and evolution in the Indo-West Pacific Ocean. Systematic Zoology, 36(3), 237–247.
- Burress, E. D., Martinez, C. M., & Wainwright, P. C. (2020). Decoupled jaws promote trophic diversity in cichlid fishes. *Evolution*, 74(5), 950–961. https://doi.org/10.1111/evo.13971
- Burress, E. D., & Muñoz, M. M. (2022). Ecological opportunity from innovation, not islands, drove the anole lizard adaptive radiation. Systematic Biology, 71(1), 93–104. https://doi.org/10.1093/ sysbio/syab031
- Castro-Santos, T., Goerig, E., He, P., & Lauder, G. V. (2022). Applied aspects of locomotion and biomechanics. In S. J. Cooke, N. A. Fangue, A. P. Farrell, C. J. Brauner, & E. J. Eliason (Eds.), Fish physiology (pp. 91–140). Academic Press.
- Chang, J., Rabosky, D. L., Smith, S. A., & Alfaro, M. E. (2019). An R package and online resource for macroevolutionary studies using the ray-finned fish tree of life. *Methods in Ecology and Evolution*, 10(7), 1118–1124. https://doi.org/10.1111/2041-210x.13182
- Clarke, A., & Crame, J. A. (2010). Evolutionary dynamics at high latitudes: Speciation and extinction in polar marine faunas. Philosophical Transactions of the Royal Society of London. Series B, Biological Sciences, 365(1558), 3655–3666. https://doi.org/10.1098/ rstb.2010.0270
- Cooper, N., & Purvis, A. (2010). Body size evolution in mammals: Complexity in tempo and mode. American Naturalist, 175(6), 727–738. https://doi.org/10.1086/652466
- Corn, K. A., Friedman, S. T., Burress, E. D., Martinez, C. M., Larouche, O., Price, S. A., & Wainwright, P. C. (2022). The rise of biting during the Cenozoic fueled reef fish body shape diversification. Proceedings of the National Academy of Sciences of the United States of America, 119(31), e2119828119. https://doi.org/10.1073/pnas.2119828119
- Deng, C., Cheng, C. -H. C., Ye, H., He, X., & Chen, L. (2010). Evolution of an antifreeze protein by neofunctionalization under escape from adaptive conflict. Proceedings of the National Academy of Sciences of the United States of America, 107, 21593–21598.
- Domenici, P., Turesson, H., Brodersen, J., & Brönmark, C. (2008). Predator-induced morphology enhances escape locomotion in crucian carp. Proceedings Biological Sciences, 275(1631), 195–201. https://doi.org/10.1098/rspb.2007.1088
- Dornburg, A., Federman, S., Lamb, A. D., Jones, C. D., & Near, T. J. (2017). Cradles and museums of Antarctic teleost biodiversity. *Nature Ecology and Evolution*, 1(9), 1379–1384. https://doi.org/10.1038/ s41559-017-0239-y
- Drummond, A. J., Ho, S. Y. W., Phillips, M. J., & Rambaut, A. (2006). Relaxed phylogenetics and dating with confidence. PLoS Biology, 4(5), e88. https://doi.org/10.1371/journal.pbio.0040088
- Drury, J. P., Clavel, J., Tobias, J. A., Rolland, J., Sheard, C., & Morlon, H. (2021). Tempo and mode of morphological evolution are decoupled from latitude in birds. PLoS Biology, 19(8), e3001270. https:// doi.org/10.1371/journal.pbio.3001270
- Eastman, J. T. (1991). Evolution and diversification of Antarctic notothenioid fishes. American Zoologist, 31(1), 93–110. https://doi. org/10.1093/icb/31.1.93
- Eastman, J. T. (2005). The nature of the diversity of Antarctic fishes. Polar Biology, 28(2), 93–107. https://doi.org/10.1007/s00300-004-0667-4

- Eastman, J. T., & DeVries, A. L. (1981). Buoyancy adaptations in a swim-bladderless Antarctic fish. Journal of Morphology, 167(1), 91–102. https://doi.org/10.1002/jmor.1051670108
- Egan, J. P., Bloom, D. D., & Simons, A. M. (2022). Time for speciation and niche conservatism explain the latitudinal diversity gradient in clupeiform fishes. *Journal of Biogeography*, 49(11), 1952– 1966. https://doi.org/10.1111/jbi.14465
- Etienne, R. S., Cabral, J. S., Hagen, O., Hartig, F., Hurlbert, A. H., Pellissier, L., Pontarp, M., & Storch, D. (2019). A minimal model for the latitudinal diversity gradient suggests a dominant role for ecological limits. *American Naturalist*, 194(5), E122–E133. https://doi.org/10.1086/705243
- Fenton, I. S., Aze, T., Farnsworth, A., Valdes, P., & Saupe, E. E. (2023). Origination of the modern-style diversity gradient 15 million years ago. Nature, 614(7949), 708–712. https://doi.org/10.1038/ s41586-023-05712-6
- Fischer, A. G. (1960). Latitudinal variations in organic diversity. Evolution, 14(1), 64–81. https://doi.org/10.2307/2405923
- Fletcher, G. L., Hew, C. L., & Davies, P. L. (2001). Antifreeze proteins of teleost fishes. Annual Review of Physiology, 63, 359–390. https:// doi.org/10.1146/annurev.physiol.63.1.359
- Fletcher, G. L., Kao, M. H., & Fourney, R. M. (1986). Antifreeze peptides confer freezing resistance to fish. Canadian Journal of Zoology, 64(9), 1897–1901. https://doi.org/10.1139/z86-284
- Folk, R. A., Stubbs, R. L., Mort, M. E., Cellinese, N., Allen, J. M., Soltis, P. S., Soltis, D. E., & Guralnick, R. P. (2019). Rates of niche and phenotype evolution lag behind diversification in a temperate radiation. Proceedings of the National Academy of Sciences of the United States of America, 116(22), 10874–10882. https://doi.org/10.1073/ pnas.1817999116
- Freeman, B. G., Weeks, T., Schluter, D., & Tobias, J. A. (2022). The latitudinal gradient in rates of evolution for bird beaks, a species interaction trait. *Ecology Letters*, 25(3), 635–646. https://doi. org/10.1111/ele.13726
- Friedman, S. T., & Muñoz, M. M. (2023). A latitudinal gradient of deep-sea invasions for marine fishes. Nature Communications, 14(1), 773. https://doi.org/10.1038/s41467-023-36501-4
- Friedman, S. T., Price, S. A., Corn, K. A., Larouche, O., Martinez, C. M., & Wainwright, P. C. (2020). Body shape diversification along the benthic-pelagic axis in marine fishes. *Proceedings of the Royal* Society B: Biological Sciences, 287(1931), 20201053. https://doi. org/10.1098/rspb.2020.1053
- Friedman, S. T., Price, S. A., Hoey, A. S., & Wainwright, P. C. (2016). Ecomorphological convergence in planktivorous surgeonfishes. Journal of Evolutionary Biology, 29(5), 965–978. https://doi. org/10.1111/jeb.12837
- Friedman, S. T., Price, S. A., & Wainwright, P. C. (2021). The effect of locomotion mode on body shape evolution in teleost fishes. Integrative Organismal Biology, 3(1), obab016. https://doi. org/10.1093/iob/obab016
- Gerringer, M. E., Linley, T. D., & Nielsen, J. G. (2021). Revision of the depth record of bony fishes with notes on hadal snailfishes (Liparidae, Scorpaeniformes) and cusk eels (Ophidiidae, Ophidiiformes). *Marine Biology*, 168(11), 167.
- Ghezelayagh, A., Harrington, R. C., Burress, E. D., Campbell, M. A., Buckner, J. C., Chakrabarty, P., Glass, J. R., McCraney, W. T., Unmack, P. J., Thacker, C. E., Alfaro, M. E., Friedman, S. T., Ludt, W. B., Cowman, P. F., Friedman, M., Price, S. A., Dornburg, A., Faircloth, B. C., Wainwright, P. C., & Near, T. J. (2022). Prolonged morphological expansion of spiny-rayed fishes following the end-Cretaceous. Nature Ecology and Evolution, 6(8), 1211–1220. https://doi.org/10.1038/s41559-022-01801-3

- Givnish, T. J. (2015). Adaptive radiation versus 'radiation' and 'explosive diversification': why conceptual distinctions are fundamental to understanding evolution. *New Phytologist*, 207(2), 297–303. https://doi.org/10.1111/nph.13482
- Gong, Z. Y., King, M. J., Fletcher, G. L., & Hew, C. L. (1995). The antifreeze protein genes of the winter flounder, *Pleuronectus americanus*, are differentially regulated in liver and nonliver tissues. Biochemical and Biophysical Research Communications, 206(1), 387– 392. https://doi.org/10.1006/bbrc.1995.1053
- Grenié, M., & Gruson, H. (2023). 2023. fundiversity: A modular R package to compute functional diversity indices. *Ecography*, 2023(3), e06585.
- Hawkins, B. A., Diniz-Filho, J. A. F., Jaramillo, C. A., & Soeller, S. A. (2006). Post-Eocene climate change, niche conservatism, and the latitudinal diversity gradient of New World birds. *Journal of Biogeography*, 33, 770–780.
- Hew, C. L., Wang, N. C., Joshi, S., Fletcher, G. L., Scott, G. K., Hayes, P. H., Buettner, B., & Davies, P. L. (1988). Multiple genes provide the basis for antifreeze protein diversity and dosage in the ocean pout, *Macrozoarces americanus*. *Journal of Biological Chemistry*, 263(24), 12049–12055.
- Hipsley, C. A., Miles, D. B., & Müller, J. (2014). Morphological disparity opposes latitudinal diversity gradient in lacertid lizards. *Biology Letters*, 10(5), 20140101. https://doi.org/10.1098/rsbl.2014.0101
- Hobbs, R. S., Hall, J. R., Graham, L. A., Davies, P. L., & Fletcher, G. L. (2020). Antifreeze protein dispersion in eelpouts and related fishes reveals migration and climate alteration within the last 20 Ma. PLoS One, 15(12), e0243273. https://doi.org/10.1371/journal.pone.0243273
- Höhna, S., Landis, M. J., Heath, T. A., Boussau, B., Lartillot, N., Moore, B. R., Huelsenbeck, J. P., & Ronquist, F. (2016). RevBayes: Bayesian phylogenetic inference using graphical models and an interactive model-specification language. Systematic Biology, 65(4), 726–736. https://doi.org/10.1093/sysbio/syw021
- Hotaling, S., Borowiec, M. L., Lins, L. S. F., Desvignes, T., & Kelley, J. L. (2021). The biogeographic history of eelpouts and related fishes: Linking phylogeny, environmental change, and patterns of dispersal in a globally distributed fish group. *Molecular Phylogenetics and Evolution*, 162, 107211. https://doi.org/10.1016/j. ympev.2021.107211
- Hubbs, C. L. (1952). Anti-tropical distribution of fishes and other organisms. Symposium on problems of bipolarity and of pan-temperate faunas. *Pacific Science Association*, 3, 324–329.
- Ingram, T. (2015). Diversification of body shape in Sebastes rockfishes of the north-east Pacific. Biological Journal of the Linnean Society, 116(4), 805–818. https://doi.org/10.1111/bij.12635
- Ingram, T., & Kai, Y. (2014). The geography of morphological convergence in the radiations of pacific sebastes rockfishes. *American Natura*list, 184(5), E115–E131. https://doi.org/10.1086/678053
- Ingram, T., & Shurin, J. B. (2009). Trait-based assembly and phylogenetic structure in northeast Pacific rockfish assemblages. *Ecology*, 90(9), 2444–2453. https://doi.org/10.1890/08-1841.1
- Kolora, S. R. R., Owens, G. L., Vazquez, J. M., Stubbs, A., Chatla, K., Jainese, C., Seeto, K., McCrea, M., Sandel, M. W., Vianna, J. A., Maslenikov, K., Bachtrog, D., Orr, J. W., Love, M., & Sudmant, P. H. (2021). Origins and evolution of extreme life span in Pacific Ocean rockfishes. *Science*, 374(6569), 842–847. https://doi. org/10.1126/science.abg5332
- Larouche, O., Benton, B., Corn, K. A., Friedman, S. T., Gross, D., Iwan, M., Kessler, B., Martinez, C. M., Rodriguez, S., Whelpley, H., Wainwright, P. C., & Price, S. A. (2020). Reef-associated fishes have more maneuverable body shapes at a macroevolutionary

scale. Coral Reefs, 39(5), 1501–1501. https://doi.org/10.1007/ s00338-020-01992-w

- Lawson, A. M., & Weir, J. T. (2014). Latitudinal gradients in climaticniche evolution accelerate trait evolution at high latitudes. Ecology Letters, 17(11), 1427–1436. https://doi.org/10.1111/ ele.12346
- Lucek, K., Keller, I., Nolte, A. W., & Seehausen, O. (2018). Distinct colonization waves underlie the diversification of the freshwater sculpin (Cottus gobio) in the Central European Alpine region. *Journal of Evolutionary Biology*, 31(9), 1254–1267. https://doi. org/10.1111/jeb.13339
- Ludt, W. B. (2021). Missing in the middle: A review of equatorially disjunct marine taxa. Frontiers in Marine Science, 8, 660984.
- Ludt, W. B., & Myers, C. E. (2021). Distinguishing between dispersal and vicariance: A novel approach using anti-tropical taxa across the fish Tree of Life. *Journal of Biogeography*, 48(3), 577–589. https://doi.org/10.1111/jbi.14021
- Martinez, C. M., Friedman, S. T., Corn, K. A., Larouche, O., Price, S. A., & Wainwright, P. C. (2021). The deep sea is a hot spot of fish body shape evolution. *Ecology Letters*, 24(9), 1788–1799. https:// doi.org/10.1111/ele.13785
- Matschiner, M., Hanel, R., & Salzburger, W. (2011). On the origin and trigger of the notothenioid adaptive radiation. *PLoS One*, 6(4), e18911. https://doi.org/10.1371/journal.pone.0018911
- May, M. R., & Moore, B. R. (2020). A Bayesian approach for inferring the impact of a discrete character on rates of continuouscharacter evolution in the presence of background-rate variation. Systematic Biology, 69(3), 530–544. https://doi.org/10.1093/ sysbio/syz069
- Miller, E. C., Hayashi, K. T., Song, D., & Wiens, J. J. (2018). Explaining the ocean's richest biodiversity hotspot and global patterns of fish diversity. *Proceedings Biological Sciences*, 285(1888), 20181314. https://doi.org/10.1098/rspb.2018.1314
- Mittelbach, G. G., Schemske, D. W., Cornell, H. V., Allen, A. P., Brown, J. M., Bush, M. B., Harrison, S. P., Hurlbert, A. H., Knowlton, N., Lessios, H. A., McCain, C. M., McCune, A. R., McDade, L. A., McPeek, M. A., Near, T. J., Price, T. D., Ricklefs, R. E., Roy, K., Sax, D. F., ... Turelli, M. (2007). Evolution and the latitudinal diversity gradient: Speciation, extinction and biogeography. *Ecology Letters*, 10(4), 315–331. https://doi.org/10.1111/j.1461-0248.2007.01020.x
- Near, T. J., Dornburg, A., Kuhn, K. L., Eastman, J. T., Pennington, J. N., Patarnello, T., Zane, L., Fernández, D. A., & Jones, C. D. (2012). Ancient climate change, antifreeze, and the evolutionary diversification of Antarctic fishes. Proceedings of the National Academy of Sciences, 109(9), 3434–3439. https://doi.org/10.1073/ pnas.1115169109
- Neat, F. C., & Campbell, N. (2013). Proliferation of elongate fishes in the deep sea. *Journal of Fish Biology*, 83(6), 1576–1591. https://doi.org/10.1111/jfb.12266
- O'Loughlin, P., Paulay, G., Davey, N., & Michonneau, F. (2011). The Antarctic region as a marine biodiversity hotspot for echinoderms: Diversity and diversification of sea cucumbers. *Deep-Sea Research*, 58, 264–275.
- Parker, E., Zapfe, K. L., Yadav, J., Frédérich, B., Jones, C. D., Economo, E. P., Federman, S., Near, T. J., & Dornburg, A. (2022). Periodic environmental disturbance drives repeated ecomorphological diversification in an adaptive radiation of Antarctic fishes. American Naturalist, 200(6), E221–E236. https://doi.org/10.1086/721373
- Pigot, A. L., Tobias, J. A., & Jetz, W. (2016). Energetic constraints on species coexistence in birds. PLoS Biology, 14(3), e1002407. https://doi.org/10.1371/journal.pbio.1002407
- Price, S. A., Friedman, S. T., Corn, K. A., Larouche, O., Brockelsby, K., Lee, A. J., Nagaraj, M., Bertrand, N. G., Danao, M., Coyne, M. C.,

Estrada, J. R., Friedman, R., Hoeft, E., Iwan, M., Gross, D., Kao, J. H., Landry, B., Linares, M. J., McGlinn, C., ... Wainwright, P. C. (2022). FishShapes v1: Functionally relevant measurements of teleost shape and size on three dimensions. *Ecology*, 103(12), e3829. https://doi.org/10.1002/ecy.3829

- Price, S. A., Friedman, S. T., Corn, K. A., Martinez, C. M., Larouche, O., & Wainwright, P. C. (2019). Building a body shape morphospace of teleostean fishes. *Integrative and Comparative Biology*, 59(3), 716–730. https://doi.org/10.1093/icb/icz115
- Price, S. A., Tavera, J. J., Near, T. J., & Peter, C. W. (2013). Elevated rates of morphological and functional diversification in reef-dwelling haemulid fishes. *Evolution*, 67, 417–428.
- Rabosky, D. L., Chang, J., Title, P. O., Cowman, P. F., Sallan, L., Friedman, M., Kaschner, K., Garilao, C., Near, T. J., Coll, M., & Alfaro, M. E. (2018). An inverse latitudinal gradient in speciation rate for marine fishes. *Nature*, 559(7714), 392–395. https://doi. org/10.1038/s41586-018-0273-1
- Randall, J. E. (1981). Anti-tropical distribution and evolution in the IndoWest Pacific Ocean. *Pacific Science*, 35, 197–209.
- Raymond, J. A., & DeVries, A. L. (1977). Adsorption inhibition as a mechanism of freezing resistance in polar fishes. Proceedings of the National Academy of Sciences, 74(6), 2589–2593. https://doi. org/10.1073/pnas.74.6.2589
- Revell, L. J. (2010). Phylogenetic signal and linear regression on species data. Methods in Ecology and Evolution, 1(4), 319–329. https:// doi.org/10.1111/j.2041-210x.2010.00044.x
- Revell, L. J. (2012). phytools: An R package for phylogenetic comparative biology (and other things). Methods in Ecology and Evolution, 3(2), 217–223. https://doi.org/10.1111/j.2041-210x.2011.00169.x
- Revell, L. J. (2013). A comment on the use of stochastic character maps to estimate evolutionary rate variation in a continuously valued trait. Systematic Biology, 62(2), 339–345. https://doi. org/10.1093/sysbio/sys084
- Reynolds, R. G., Collar, D. C., Pasachnik, S. A., Niemiller, M. L., Puente-Rolón, A. R., & Revell, L. J. (2016). Ecological specialization and morphological diversification in Greater Antillean boas. *Evolution*, 70(8), 1882–1895. https://doi.org/10.1111/evo.12987
- Ricklefs, R. E. (2012). Species richness and morphological diversity of passerine birds. Proceedings of the National Academy of Sciences, 109(36), 14482–14487. https://doi.org/10.1073/pnas.1212079109
- Roy, K., & Martien, K. K. (2001). Latitudinal distribution of body size in north-eastern Pacific marine bivalves. *Journal of Biogeography*, 28(4), 485–493. https://doi.org/10.1046/j.1365-2699.2001.00561.x
- Rutschmann, S., Matschiner, M., Damerau, M., Muschick, M., Lehmann, M. F., Hanel, R., & Salzburger, W. (2011). Parallel ecological diversification in Antarctic notothenioid fishes as evidence for adaptive radiation. *Molecular Ecology*, 20(22), 4707– 4721. https://doi.org/10.1111/j.1365-294X.2011.05279.x
- Salisbury, C. L., Seddon, N., Cooney, C. R., & Tobias, J. A. (2012). The latitudinal gradient in dispersal constraints: Ecological specialisation drives diversification in tropical birds. *Ecology Letters*, 15(8), 847–855. https://doi.org/10.1111/j.1461-0248.2012.01806.x
- Schluter, D. (2000). The ecology of adaptive radiation. Oxford University Press.
- Schluter, D. (2016). Speciation, ecological opportunity, and latitude. American Naturalist, 187(1), 1–18. https://doi.org/10.1086/684193
- Schluter, D., & Pennell, M. W. (2017). Speciation gradients and the distribution of biodiversity. Nature, 546(7656), 48–55. https://doi. org/10.1038/nature22897
- Simpson, G. G. (1944). Tempo and mode in evolution. Columbia University Press.
- Simpson, G. G. (1953). The major features of evolution. Columbia University Press.

- Stuart-Smith, R. D., Bates, A. E., Lefcheck, J. S., Duffy, J. E., Baker, S. C., Thomson, R. J., Stuart-Smith, J. F., Hill, N. A., Kininmonth, S. J., Airoldi, L., Becerro, M. A., Campbell, S. J., Dawson, T. P., Navarrete, S. A., Soler, G. A., Strain, E. M. A., Willis, T. J., & Edgar, G. J. (2013). Integrating abundance and functional traits reveals new global hotspots of fish diversity. *Nature*, 501(7468), 539–542. https://doi. org/10.1038/nature12529
- Tytell, E. D., & Lauder, G. V. (2004). The hydrodynamics of eel swimming: I. Wake structure. *Journal of Experimental Biology*, 207(Pt 11), 1825–1841. https://doi.org/10.1242/jeb.00968
- van Ginneken, V., Antonissen, E., Müller, U. K., Booms, R., Eding, E., Verreth, J., & van den Thillart, G. (2005). Eel migration to the Sargasso: Remarkably high swimming efficiency and low energy costs. Journal of Experimental Biology, 208, 1329–1335.
- Villéger, S., Mason, N. W. H., & Mouillot, D. (2008). New multidimensional functional diversity indices for a multifaceted framework in functional ecology. Ecology, 89(8), 2290–2301. https://doi. org/10.1890/07-1206.1
- Wainwright, P. C., & Longo, S. J. (2017). Functional innovations and the conquest of the oceans by acanthomorph fishes. *Current Biology*, 27(11), R550–R557. https://doi.org/10.1016/j.cub.2017.03.044

- Webb, P. W. (1982). Locomotor patterns in the evolution of actinopterygian fishes. American Zoologist, 22(2), 329–342. https://doi. org/10.1093/icb/22.2.329
- Webb, P. W. (1983). Speed, acceleration and manoeuvrability of two teleost fishes. Journal of Experimental Biology, 102(1), 115–122. https://doi.org/10.1242/jeb.102.1.115
- Webb, P.W. (1984). Form and function in fish swimming. *Scientific American*, 251(1), 72–82. https://doi.org/10.1038/scientificamerican0784-72
- Webb, P. W. (1988). Simple physical principles and vertebrate aquatic locomotion. American Zoologist, 28(2), 709–725. https://doi. org/10.1093/icb/28.2.709
- Weir, J. T., & Wheatcroft, D. (2010). A latitudinal gradient in rates of evolution of avian syllable diversity and song length. Proceedings of the Royal Society B: Biological Sciences, 278(1712), 1713–1720. https://doi.org/10.1098/rspb.2010.2037
- Wiens, J. J. (2023). Trait-based species richness: Ecology and macroevolution. Biological Reviews of the Cambridge Philosophical Society, 98(4), 1365–1387. https://doi.org/10.1111/brv.12957
- Wiens, J. J., & Donoghue, M. J. (2004). Historical biogeography, ecology and species richness. *Trends in Ecology and Evolution*, 19(12), 639–644. https://doi.org/10.1016/j.tree.2004.09.011. Elsevier